**Author’s response to reviews**

**Title:** Three novel mutations in the ATP7B gene of unrelated Vietnamese patients with Wilson disease

**Authors:**
Nguyen Huy Hoang (nhhoang@igr.ac.vn)
Nguyen Huong (nguyenmaihuong@nch.org.vn)
Nguyen Lien (ntkimlienibt@gmail.com)
Ngo Ngoc (ngodiemngoc@gmail.com)
Nguyen Mai (nguyenphuongmai@nch.org.vn)
Nguyen Hoa (dranhhoa@gmail.com)
Le Hai (hailt@nhp.org.vn)
Phan Chi (pvchi@yahoo.com)
Ta Van (tathanhvan@hmu.edu.vn)
Tran Khanh (Khanhvan7364@gmail.com)

**Version:** 6 **Date:** 24 May 2018

**Author’s response to reviews:**

MGTC-D-18-00005R5

Three novel mutations in the ATP7B gene of unrelated Vietnamese patients with Wilson disease

Nguyen Huy Hoang; Nguyen Thi Mai Huong; Nguyen Thi Kim Lien; Ngo Diem Ngoc; Nguyen Thi Phuong Mai; Nguyen Pham Anh Hoa; Le Thanh Hai; Phan Van Chi; Ta Thanh Van; Tran Van Khanh

BMC Medical Genetics
May 25, 2018

Dear Prof. Editors,

Editor of BMC Medical Genetics

Thank you for your letter on May 23, 2018.

Please find enclosed the revised version of our manuscript MGTC-D-18-00005R5. According to the highly valuable comments of the Editors, we modified the manuscript:

1. Please move this text "The study was approved by the Scientific Committee of Institute of Genome Research under reference number 12/QD-NCHG." from the Consent for publication section to the Ethics approval and consent to participate section.

This sentence has been moved to the Ethics approval and consent to participate section.

2. Please put all information regarding participants consent for publication in the Consent for publication section. We will not be able to publish your manuscript until this is completed correctly.

All information regarding participants consent for publication have been put in the Consent for publication section.

3. Please remove the second Conclusions heading, above the sentence "In our study, we identified five nucleotide".

The second Conclusions heading has been removed.

4. Please remove the attached Ethics approval form and consent forms. You must include a statement in the Consent for publication section stating that consent for publication was obtained, that it was written, who it was obtained from (e.g. patients/parents), and what they consented to specifically, e.g. medical information, images, videos, etc.

The information for Consent for publication have been written in this section.
5. Please remove the funding information from the Acknowledgements and include it in the Funding section instead.

This section has been written as suggested by the Editors.

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

The manuscript will be uploaded as a single, final, clean version as suggested by the Editors.

Once again thank you very much.

Sincerely Yours,

Nguyen Huy Hoang